

The current issue and full text archive of this journal is available at www.emeraldinsight.com/0960-0035.htm

IJPDLM 43,5/6

380

Received 23 March 2012 Revised 23 August 2012 3 October 2012 Accepted 10 October 2012

# Consumer preferences for sustainability and their impact on supply chain management The case of mobile phones

Anu Bask

Academy of Finland and Aalto University School of Business, Helsinki, Finland, and

Merja Halme, Markku Kallio and Markku Kuula Department of Information and Service Economy, Aalto University School of Business, Helsinki, Finland

## Abstract

**Purpose** – Consumer values increasingly favor sustainable development in products and services, thereby fostering the need to develop new operational and managerial practices that support sustainability in supply chain management. The purpose of this study is to identify relevant product features related to sustainable development in this context, and use the choice of mobile phone as an example in measuring their importance.

**Design/methodology/approach** – The study used two different methods (qualitative and quantitative) in two phases. First it organized focus-group discussions in order to identify the features of sustainability that affect the choice of a mobile phone. The most significant features served as a starting point for the choice of attributes to be included in the final step, choice-based conjoint analysis (CBC), which assesses respondents' value functions by means of latent class clustering. Between the two major phases it carried out two additional pre-tests in order to reduce the number of attributes.

**Findings** – The results provide fundamental information concerning the relative importance of sustainability features in the selection of a mobile phone. The study identified four different clusters of purchasers: updaters, budgeters, environmentalists, and long-life users. According to the findings, some consumers are willing to pay a premium for sustainability features. The authors discuss the potential implications of the results in the context of supply chain design.

**Originality/value** – The literature on supply chain management tends to see the consumers as a "black box". This paper reports the first results of opening this box by linking the supply chain perspective to consumer choice behavior.

Keywords Sustainability, Supply chains, Consumer preferences, Mobile phones, Supply chain management, Consumer behavior, Finland

Paper type Research paper

International Journal of Physical Distribution & Logistics Management Vol. 43 No. 56, 2013 pp. 380-406 © Emerald Group Publishing Limited 0960-0035 DOI 10.1108/JJPDLM-03-2012-0081

The authors are grateful to the Guest Editor and four anonymous reviewers for their valuable comments and suggestions for improving the paper. They would also like to thank the Jenny and Antti Wihuri Foundation for financial support, TNS Gallup Finland for allowing them to use five statements from their Atlas study and providing them with data, and the Aalto University Service Factory for their encouragement of interdisciplinary research.



# 1. Introduction

Environmental and social issues have played a prominent role in industrial countries during the last few decades. Various green movements have emerged, and new concepts such as the greenhouse phenomenon, acid rain, and social conscience are commonly used in everyday spoken language. As global awareness of environmental problems has grown, consumers have become the crucial factor in promoting green designs (Schischke et al., 2005). Thus, consumers have become more aware of the consequences of their consumption decisions and their choices are increasingly affecting the product offerings (Andersen and Skjoett-Larsen, 2009; Defee et al., 2009). Consumers are no longer solely interested in the physical product, such as the materials used for production and pacing, but also wish to know, for instance, where the raw materials were produced and purchased. In addition, many consumers take an interest in what happens to the product after its lifetime. In other words they are increasingly interested in the social and environmental impact of the entire supply chain (SC). As for the companies, they have been under increasing pressure to consider the environmental consequences of their products and services during the past 20 years (Kleindorfer *et al.*, 2005). It is becoming more and more important to be aware of the implications of supply-chain decisions for the life cycle of the product.

Concern with sustainable development has also spread within SCs and their management (Fabbe-Costes et al., 2011; Seuring and Muller, 2008). Nowadays, sustainability is seen as a source of competitive advantage, and even as the backbone of innovation (Nidumolu et al., 2009). The political environment has changed in industrial countries, for example, and governments have introduced new and increasingly strict laws and international standards covering environmental aspects (Lee and Kim, 2009). Focal companies therefore need to co-operate with other parties in SC networks, and also with governments and regulatory bodies (Fabbe-Costes et al., 2011). They should know where their suppliers stand on aspects of sustainable development (Seuring and Muller, 2008). Naturally, manufacturers of mobile phones are among those putting great effort into sustainability and associating themselves with its values. The results are visible in the SAM Corporate Sustainability Assessment Report, for example, Nokia being awarded the SAM Silver Class and Motorola the Bronze Class (The Sustainability Yearbook, 2012). Moreover, several proactive manufacturers, such as Nokia, Samsung, and Sony Ericsson, comply globally with EU regulations, which are the most comprehensive in existence (Shiloy, 2009). Furthermore, some manufacturers have introduced a variety of green phones (Zadok and Puustinen, 2010; Bask and Kuula, 2011), and openly share information on environmental and social issues that affect the entire SC.

In spite of these efforts aimed at sustainable development, there is a lack of academic literature on how consumers view or value the sustainable features of a product or service, and even fewer studies take a SC perspective on sustainability and link it to consumer choice behavior. Our aim in this paper is to narrow this research gap, focusing on the case of mobile phones, and to study consumer preferences for sustainability features from a SC perspective. We also discuss what, if any, implications our findings on consumer preferences might have for SC design. We had several reasons for choosing the mobile phone as our object of analysis, the prime ones being that it is a modular, high-involvement product (contrary to those most often studied in the "green product" context), it is global in usage with huge volumes, and its role is expected to grow in both Western and developing countries. The number of mobile-phone users is expected



Preferences for sustainability

to reach as high as around six billion by the year 2013 (GSMA, 2010): by way of comparison, global sales of mobile phones reached 1.2 billion in 2008, of which 229 million units were sold in the biggest market area, the EMEA region (Gartner, 2009). Moreover, short life cycles are the norm nowadays (Hilletoft *et al.*, 2010), and this is the case with mobile phones, which require repeat-purchase decisions. From the production perspective mobile phones are modular, which facilitates the fast adoption of new innovations. On the global level, their importance is expected to increase in the future as mobile technology develops to embrace numerous services: they will take on the roles of credit cards, online marketing information receivers and browsers, and televisions, for example (Lyytinen *et al.*, 2004). Mobile technology has already showed great promise in addressing health problems in the third world (Lynn, 2010; Kaplan, 2006) and in the area of payment services (e.g. M-PESA in Kenya, Intuit 2020 Report, 2010).

Therefore, at the same time as volumes are growing, sustainability issues are becoming increasingly important with regard to mobile phones and their SCs. Our objective in this study is to measure the relative importance for the consumer of the manifold facets of sustainability in a high-involvement product, taking the whole SC as a starting point: this has thus far not been done in academic studies. The key research question is this: from the consumer's point of view, what are the most important sustainability features in SCs with regard to mobile phones?

The rest of the article is organized as follows. Sections 2 and 3 review the literature on sustainability in SC management, and consumer choice behavior, respectively. Section 4 describes the methods used, Section 5 presents the results of the choice-based conjoint (CBC) analysis, and Section 6 discusses the potential implications for SC management. Section 7 concludes the paper.

#### 2. Sustainability in supply chain management

The holistic view on sustainable supply chain management (SSCM) covers environmental, economic and social aspects. Carter and Rogers (2008) define SSCM as:

[...] strategic, transparent integration and achievement of an organization's social, environmental, and economic goals in the systemic coordination of key interorganizational business processes for improving the long-term economic performance of the individual company and its supply chains.

According to Carter and Easton (2011), SSCM relates to the long-term improvement of organizations and has implications for companies' economic bottom lines. Arguing that engaging in SSCM is a requirement for successful business, they provide a framework that describes the relationships among these triple bottom-line aspects and assigns supply-chain professionals key roles in terms of implementing sustainable practices. The increasing significance of SSCM has arisen from the need to find new operational and managerial practices to reduce the environmental impact of the operations of individual companies and entire SCs. For example, cost pressures based on frequent changes in commodity prices, limited supplies of critical materials, and long transportation distances have forced companies to question and develop their sourcing and SCM strategies. Therefore, development issues in sustainability involve not only firms but also entire SCs (Fabbe-Costes *et al.*, 2011). It is therefore of high interest to identify environmental and social initiatives that have had the greatest economic impact on SCs (Carter and Jennings, 2002), and research on corporate social responsibility (CSR) is increasingly



**IIPDLM** 

43,5/6

linked to SSCM. Thus, the role of CSR has changed in that it also currently encompasses entire SCs rather than individual company domains (Andersen and Skjoett-Larsen, 2009; Spence and Bourlakis, 2009). Furthermore, Lee and Kim (2009) point out that companies embracing corporate responsibility recognize that environmental impacts need to be managed in the same way as financial and commercial performance. They also suggest that if companies wish to be competitive and survive in the markets, they need to extend their CSR activities to the whole SC.

Researchers have used different terms and definitions of environmental sustainability. Shang *et al.* (2010) and Srivastava (2008) adopt the term green supply chain management (GSCM)). Srivastava (2008) defines GSCM as:

Integrating environmental thinking into supply chain management including product design, material sourcing and selection, manufacturing processes, delivery of the final product to the consumers as well as end-of-life management of the product after its useful life.

Defee *et al.* (2009) use the term closed-loop supply chain orientation (CLSCO), which they define rather similarly: "Recognition of the strategic and environmental implications of activities and processes involved in managing and integrating the forward and reverse flows of closed-loop supply chains". According to Defee *et al.* (2009), CLSCO is a prerequisite for effectively leveraging closed-loop supply chains (CLSC) capabilities. Guide (2000) and Andersen and Skjoett-Larsen (2009) use the term CLSC in the context of shortened product life-cycles and increased needs to recover assets through recycling and reuse of materials and components. According to Kleindorfer *et al.* (2005), closing the loop in SCs should change the business models in use. In short, all the approaches to sustainability discussed above assign SCM a strategic rather than an operational role.

It is obvious from the above that the concepts of sustainable development are increasingly being linked with product design, sourcing and materials management, and with the reduction of resources and waste in SCs (Shang *et al.*, 2010; Dangelio and Pontrandolfo, 2010; Srivastava, 2008; Linton *et al.*, 2007; Fiksel, 2001; Luttropp and Lagerstedt, 2006; Stahel, 2001). A sustainable SC could therefore be built on the following pillars: sustainable product design, sustainable sourcing and manufacturing processes, as well as reverse logistics operations and coordination. From the mobile phone manufacturer's point-of-view there are several aspects that can be used to operationalize SSCM issues. Therefore, for further analysis of SSCM a four-theme framework was developed based on:

- (1) sustainable strategy and policy;
- (2) sustainable product design;
- (3) sustainable sourcing; and
- (4) end-of-life management.

We drew up statements to characterize the different facets of these four areas, and they served as a starting point for our semi-structured focus group discussions (users' perspective).

First, the sustainable strategy and policy issue when related to a manufacturer includes such aspects as: the company has a vision/strategy/and/or policy regarding sustainability in use (Angell and Klassen, 1999); it has a structured environmental assessment tool (e.g. life-cycle assessment (LCA)) in place (Hervani *et al.*, 2005); it invests in waste, energy and emissions management; it publishes a CSR report



Preferences for sustainability (Chen and Bouvain, 2009); and it participates in global development initiatives. Life-cycle management is an important part of SSCM, and LCA facilitates the structured evaluation of the total environmental load (Hervani *et al.*, 2005), product life cycles, and supply-chain aspects. The benefit is that the company can direct the focus of green designs to product and process innovations (Angell and Klassen, 1999), and can more easily adopt SSCM principles (Darnall *et al.*, 2008).

Second, given that "a world-class supply chain starts with product design" (Stank et al., 2011), sustainable product design brings sustainability aspects to the whole SC from product and production planning to end-of-life management. In fact, this stage determines 80 percent of the SC's costs (Carter and Ellram, 1998; Stank et al., 2011) and thus affects various stages such as disassembly, reuse, and recycling (Carter and Jennings, 2002). Hilletoft et al. (2010) emphasize that new product development (NPD) should be driven by customer expectations, which calls for profound knowledge of them. Customer initiatives, perceptions and opinions could therefore have implications for SCM design, thus there is a need for the continuous collection of information on consumer needs and benefits in order to tailor SCs to target segments (Hilletoft *et al.*, 2010). According to Hilletoft *et al.* (2010), companies should work simultaneously with customer-oriented NPD and SCM design. The sustainable product design in the context of mobile phones include such aspects as: favoring renewable and recvclable resources (Tischner, 2001), incorporating recycling and disassembly (dismantling) options (Defee et al., 2009; Andersen and Skjoett-Larsen, 2009; Kleindorfer et al., 2005), imposing stricter requirements related to environmentally hazardous materials than stipulated in EU regulations for example, and using analysis tools for materials certified by a third party (CSR criteria; Tischner, 2001).

Third with regard to sustainable sourcing, as companies increasingly rely on their suppliers in order to attain competitive advantage (Handfield *et al.*, 2005), in today's globally networked world principles of sustainability should prevail. Given the increasing amount of outsourcing from developing and low-cost countries, concerns about the social and environmental impact have risen (Andersen and Skjoett-Larsen, 2009). Sustainable sourcing includes such aspects as: the extent to which suppliers engage in environmental initiatives (e.g. recycling, hazardous material, and waste processing in production), develop their operations, and ensure safe and ethical working environments at their plants (Carter, 2005; Carter and Jennings, 2002). On the operational level the manufacturer should: have a certified tool for evaluating its suppliers' environmental sustainability (Hervani *et al.*, 2005), systematically monitor and audit its suppliers (Spence and Bourlakis, 2009), train its suppliers in sustainability issues (Lee and Kim, 2009), systematically share information regarding materials that require control (Handfield *et al.*, 2005), and provide clear instructions regarding the use of resources and waste management (Scott, 2008).

Fourth, product recovery and end-of-life management entail the recycling, re-use and remanufacturing of products and materials (Daugherty *et al.*, 2001; Rogers and Tibben-Lembke, 2001; Stock, 1998). End-of-life management policy covers recycling, and developing efficient processes for handling returns and product take-backs, which are examples of putting sustainability-oriented strategy into practice. In the context of mobile phones, aspects of product recovery and end-of-life managements are such as: stricter practices in the reuse and recycling of materials than required under EU regulations, participation in programs aimed at product recycling, re-selling of the



IJPDLM

43,5/6

products that have been recycled (Tischner, 2001; Guide and van Wassenhove, 2002), repairing and upgrading products that have been recycled or are outdated (Tischner, 2001; Defee *et al.*, 2009), reselling them, and sharing of information on the use of recycled materials and components.

Several authors point out that the world-class design of a SC starts with the customer (Stank *et al.*, 2011). Nevertheless, very few studies have attempted to decompose customer perceptions of sustainability into an operational set of features. Thus, the SCM literature typically regards the customer as a "black box". The aim in this study, therefore, is to link the features of sustainability as consumers see them with aspects of SCs. Next we briefly review the literature on sustainability and consumer choice.

#### 3. Consumer choice behavior

Chen (2001) acknowledges that "greening itself is not a well-defined concept", meaning that consumers, manufacturers, dealers and governments view the benefits of a product/service in different ways. His list of typical environmental attributes includes recyclability, recycled content, fuel efficiency, toxic content reduction and emission-related performance. He does not discuss matters related to social corporate responsibility, however. It is not at all obvious which facets of sustainable development bring most value to consumers, and the features are very likely to vary across different products/services.

Various studies assess the importance of sustainable development in consumers' purchase decisions. The choice of the attributes included in the analyses is based on experience or previous literature, and the effect of eco-labels, for example, is also considered. The product categories covered are limited mainly to food (Moon et al., 2002; Ubilava et al., 2010; van Doorn and Verhoef, 2011; Rokka and Uusitalo, 2008), other low-involvement groceries (Bjorner et al., 2004; Auger et al., 2003), energy (Roe et al., 2001), and wood-based furniture (Veisten, 2007). Train and tourist services have also been studied (Eggers and Eggers, 2011; Hearne and Salinas, 2002). Among high-involvement products vehicles (Potoglou and Kanaroglou, 2007) and household appliances have been objects of interest (Sammer and Wustenhagen, 2006; Sonnenberg et al., 2011). Potoglou and Kanaroglou (2007) considered preferences for gasoline, alternative fuelled and hybrid cars with regard to one sustainable-development attribute, "pollution level". Sammer and Wustenhagen (2006) exceptionally included three attributes in their study on washing machines: water consumption, energy consumption, and energy-efficiency rating. Sonnenberg et al. (2011) used direct questioning in assessing the importance of several types of sustainability information in the purchase of a household appliance.

The most commonly used assessment methods include CBC analysis and discrete choice analysis. Direct questioning has also been employed (Sonnenberg *et al.*, 2011; Veisten, 2007; Moon *et al.*, 2002), as well as the analysis of scanner panel data (Bjorner *et al.*, 2004). Aggregate results or results based on some demographic category such as country or income are commonly reported, but in some cases cluster analysis is used in order to give a more detailed view of various attitudes towards features of sustainable development. Many studies attempt to assess consumers' willingness to pay for a particular feature – and here we mention only some examples: Veisten (2007) found the premium to be in the range of 2-16 percent paid for eco-labeled wood furniture, the premium in eco-food was 28-45 percent (Moon *et al.*, 2002; Ubilava *et al.*, 2010),



Preferences for sustainability

385

and 30 percent in a high-involvement product such as a washing machine (Sammer and Wustenhagen, 2006). These figures were based on conjoint analysis (CA) or direct questioning. The marginal willingness to pay for the Nordic Swan eco-label in toilet paper was found to be 13-18 percent, the calculations being based on the purchases of a household panel in Denmark. A real choice experiment related to the purchase of roses revealed a premium exceeding the rose price by more than 100 percent (Michaud *et al.*, 2012) – here the real choice indicates that real roses and actual transactions against money featured in the experiment.

Thus, the existing research indicates that premiums for sustainable features exist. However, several studies also suggest that even if consumers have positive attitudes toward sustainability, there seems to be a gap between attitudes and actions (Luchs *et al.*, 2010; Hughner *et al.*, 2007). Thus, the results of studies that did not include real transactions and economic consequences should be viewed with care.

Several studies also attempt to identify the determinants of green purchase behavior (Laroche *et al.*, 2001; Chan, 2001; Ubilava *et al.*, 2010) in order to target consumers with the most favorable attitudes towards paying a premium for sustainable-development attributes. We are confident that without a major risk of pro-innovation bias (Rogers, 1976), the number of such consumers will grow as consciousness of the need for sustainable development increases. Today these consumers are the innovators (or early adopters) in the diffusion of innovations life-cycle (Rogers, 1962).

#### 4. Methods

The objective of the study was to find out to what features of sustainable development, covering the whole SC, consumers assign value when choosing a mobile phone. We were also interested in the possible impacts of these features on the design of the chain. The key question is this:

*Q1.* What are the features of SC sustainability that consumers shopping for mobile phones value most?

We also addressed three additional questions:

- *Q2.* Is it possible to identify consumer groups with different preference structures, and if so, what are they?
- Q3. Are consumers willing to pay more for a mobile phone with features promoting sustainable development?
- *Q4.* What are the potential implications of the identified consumer preferences for SC management?

In the absence of previous studies that could be used as a starting point we started by conducting interviews in order to gather information about the features that play a crucial role in consumer choice with regard to mobile phones. The final goal was to measure quantitatively the relative importance of the different features of sustainable development.

The study had two major phases. In the first we decided to use focus groups in order to identify the most significant sustainable features of mobile phones, and in the second phase we assessed the importance of these features by means of CBC analysis. Beginning with focus group discussions is particularly well suited to studies in which



**IIPDLM** 

43,5/6

the aim is to discover factors that influence behavior, motivation, opinions, or feelings and ideas that people may have about something (Krueger and Casey, 2000; Morgan, 1996). The strength of the focus-group method lies in the interactive discussions in that participants are free to ask each other questions, and to reconsider and re-evaluate their own understanding of their experiences or opinions as the information generated during the interaction accumulates (Halcomb *et al.*, 2007; Morgan, 1996, 1998; Gibbs, 1997; Kitzinger, 1995). We therefore considered it appropriate in addressing Q1.

We chose CA for the measurement as it has a long track record of preference measurement, initially primarily in marketing research but currently also in areas such as health care, food studies, and transportation, and it facilitates the estimation of value functions on the segment or even the individual level. CBC analysis is currently rather extensively applied in the field of operations management and SCs (Karniouchina *et al.*, 2009). Reutterer and Kotzab (2000), for example, used CA to evaluate the expectations and values of SC managers related to the design of the SC.

We interviewed a sample of Finnish university students representing three disciplines. We had one pretest group, and additionally conducted one focus group discussion per discipline. Given the students' high educational level, they could be expected to be conscious innovators or early adopters of sustainable-development aspects in their choices: in fact, the results of our study supported this assumption (see Section 5.1 and the comparison of attitudes towards sustainable development among our sample population compared with a sample representing the Finnish population as a whole). Many employees in Finland receive their mobile phones through their jobs and therefore may never make a purchasing decision. Students, in turn, are likely to lack this benefit, and typically buy their phones themselves. Thus, they are bound to evaluate and rate factors affecting their purchasing decisions. They probably also are better informed and give more attention to sustainability than the average person. Furthermore, they are likely to be decision makers and opinion leaders in the future, and may even influence their future employers in their choice of mobile phone. For these reasons we believe our choice of a student sample was iustified.

The focus-group discussions identified various features of sustainable development (Section 4.1), which we further refined and summarized as we deepened our understanding of the major facets of the phenomenon. We then conducted two additional pretests (Section 4.4) in order to further synthesize the features. The final CBC questionnaire thus included five features and the price attribute. We used CBC analysis and subsequently Latent Class clustering in addressing Q2 and Q3. Reviewing these clusters we found we were able to identify potential implications for SC management, thereby answering Q4. We then considered these implications in the light of SC design (Section 6). Next we briefly report on the focus-group discussions and then move on to the CA.

#### 4.1 Focus group discussions

The focus group discussions aimed at identifying the potential facets of sustainable development in SCs that might affect consumers' choice of mobile phones. The student sample represented three disciplines, economics and business administration, technology, and art and design. We arranged four semi-structured discussions in which a total of 22 students participated. Each interview lasted between



Preferences for sustainability

387

IJPDLM 43,5/6

388

two and three hours, during which time some snacks were served. The first group (comprising five members) was used as a test group, the aim being to generate a very thorough and holistic discussion about sustainability aspects in mobile phone SCs. The members were therefore doctoral students, who could be expected to be analytical and motivated, capable of improving and refining the elements that made up the four-theme framework (Section 2). The final statements employed in the following three focus groups with graduate students are given in the Appendix.

All the focus group discussions were semi-structured in nature, and followed the same agreed schedule. The students were first requested independently to come up with features of sustainability related to mobile phones. They were given post-its, and were asked to write each feature on a separate paper. There was no limit on the number of features. Thereafter each one was assigned to one of the four issues in the four-theme framework: sustainable strategy and policy, sustainable product design, sustainable sourcing, and sustainable end-of-life management. The respondents were then given the list of statements in the four-theme framework and were asked to accept, delete, redevelop and add new features as appropriate. After that the modified statement lists were printed for each group member. The individual members were asked to mark the statements they found relevant, and then to mark the ten most significant ones. Finally, they rated their chosen ten statements on a scale ranging from 1 (slightly important) to 3 (highly important). The most significant features with the highest total ratings across all the four groups served as a starting point for the choice of attributes to be included in the CA.

#### 4.2 Conjoint analysis

According to multi-attribute utility theory the utility of a product/service is a function of its attributes. In CA the total perceived utility U is a sum of the total value V and a random error term  $\varepsilon$  present in the valuation of a product:  $U = V + \varepsilon$ .

The total value *V* is often considered to be an additive function of product attributes, possibly with interaction terms. Each attribute is assumed to have only a finite set of possible values, called levels. If there are *n* attributes with levels denoted by  $a_1, a_2, \ldots, a_n$  the total utility *U* is:

$$U = u_1(a_1) + \dots + u_n(a_n) + \sum_{i,j} u_{ij}(a_i, a_j) + \varepsilon,$$
(1)

where  $u_i$  is a utility function of an individual attribute i, i = 1, ..., n, and  $u_{ij}$ , for some i and j, is an interaction term. The estimated values of  $u_i(a_i)$  are part-worths or partial utilities of the attribute levels, and  $u_{ij}(a_i, a_j)$  are attribute interactions. CA uses respondent preferences to estimate the respondents' value functions. In CBC analysis the preferences are elicited so that the respondent faces a number of tasks in which product profiles described in terms of attribute levels are presented, among which the respondent indicates the best one. The value functions can be estimated on the aggregate, segment or individual level.

In CBC analysis the error term  $\varepsilon$  has a Gumbel distribution (with a location parameter of zero and variance II/6). The evaluation of alternative profiles, with independent valuation errors, results in the multinomial logit choice model (McFadden, 1974; Swait and Louviere, 1993): if  $\kappa$  product profiles with total values  $V_1, \ldots, V_{\kappa}$  are compared, the probability  $p_l$  that profile l is chosen is:



$$p_l = \frac{e^{V_l}}{\sum_{s=1}^{\kappa} e^{V_s}}.$$

The following phases describe the CA process: choice of relevant product attributes and attribute levels, choice of data-gathering method, choice of utility function, choice of questionnaire parameters, and analysis of the value functions.

#### 4.3 Latent class clustering

The estimation of the segment-wise value functions was carried out using the Latent Class 4.0.8 (Sawtooth Software) clustering for choice data (DeSarbo *et al.*, 1995). In the latent class estimation, given the number of clusters, both the sizes and the cluster-wise value functions were optimized using the maximum likelihood criterion. For each respondent the probability of belonging to each cluster (membership) is also subject to optimization. As the likelihood problem is non-convex each run does not result in a guaranteed global optimum. Therefore, for each pre-set number of clusters, several repetitive runs are made and the best solution (in terms of likelihood) among the alternatives is chosen. The choice of the number of clusters is based upon the CAIC measure defined by:

CAIC = -2(Log Likelihood) + (tr + r - 1)(ln N + 1),

where r is the number of clusters, t is the number of independent parameters estimated per group and N is the total number of choice tasks in the data set. DeSarbo *et al.* (1995) suggest that the number of clusters with minimum CAIC should be used provided that it is "managerially interpretable". They also suggest using an entropy-based measure (Ramaswamy *et al.*, 1993) as well in order to determine "whether the derived segments are well separated".

#### 4.4 Questionnaire

Next we describe the steps in devising the conjoint questionnaire.

*Choice of relevant product attributes and attribute levels.* In CA features are always called attributes, and their possible values are referred to as levels. In the following we use the word attribute when we mean features to be included in the CA. On the basis of the focus group results we carried out an additional pre-test in order to reduce the number of attributes. The pre-test was conducted by means of a web-based questionnaire and maximum difference scaling (Finn and Louviere, 1992), on a convenience sample of 72 respondents. The number of attributes pre-tested was 13, and on the basis of the results some were omitted given their relatively minor importance. Moreover, some related features were summarized into one attribute. At this stage price was also included as an attribute in order to assess consumers' willingness to pay for sustainable features. We assumed a 10 percent increase in the current price of the phone, given that the focus-group as well as the first pre-test participants had assessed their willingness to pay (in relative terms) at an average of below 15 percent.

Subsequently, we tested several versions of the questionnaire comprising six sustainable-development features plus the price attribute, and improved it iteratively in line with comments we received. Thus, in the last stage, the research-group members personally interviewed a total of 21 respondents in depth, recorded their feedback in detail and discussed it. At this stage one additional sustainability attribute was



389

Preferences for sustainability

IJPDLM 43,5/6 dropped owing to its minor relevance to the respondents compared with the remaining attributes: this was suspected in earlier stages. Moreover, we made some improvements to the text and the appearance before fielding the final version. Each attribute in the final set of five combined several features dealt with in the focus group. The final sustainable-development attributes were:

390

- (1) physical strength and length of life;
- (2) updating characteristics: software and hardware;
- (3) recycling (plastic and metal parts and phone recycling);
- (4) hazardous materials and waste processing in production; and
- (5) ethical labor aspects and environmental conditions with suppliers.

Each attribute was chosen to have only two possible levels appearing in the profiles. The reference mobile phone was one that was available at the time of study, of which durability and length of life as well as updating characteristics were on the same level as current mobiles in the stores ("same as current"). There was no information about the recycling, hazardous materials and waste processing in production, or about ethical labor aspects and environmental conditions with suppliers, and the price was the current price. For all the sustainable-development attributes the more preferred level was "improved compared to the current situation". For price the two levels were, "same as the current price" and "10 percent higher than the current price". In the questionnaire the status quo levels of the attributes where linked with the grey color and the improved levels of the sustainable-development attribute were indicated by the green color. The 10 percent increase in price was highlighted in a shade of orange. Please look at Figure 1 for the attributes and their levels.

*Choice of data-gathering method and utility function.* We used a web-based questionnaire, because it was an efficient and convenient way to reach the respondents in the sample. The additive utility function with interactions was used.

*Choice of questionnaire parameters.* The questions were thus tasks in which the respondent chose the best one among the product offered. We used ten generated tasks and two fixed tasks for each respondent, the latter being used as holdouts. The number of product profiles displayed in each task was three. We did not use more profiles owing to the relatively large number of attributes. An example of a task in the questionnaire can be seen in the Figure 1.

We used SSI Web 7.0.22 with the option "balanced overlap" in generating the tasks. That resulted in a random experimental question design, which was close to orthogonal (Chrzan and Orme, 2000). There were 999 different versions of the questionnaire.

## 5. The sample and the results

#### 5.1 The sample of respondents

We sent the questionnaire to a total of 8,592 student e-mail addresses, of which 172 messages could not be delivered. All the students in the School of Economics as well as the University of Art and Design were invited to take part. Systematic sampling was used in recruiting students from the University of Technology due to the large student population. A total 1,398 completed responses were received. Table II gives the sample characteristics. The questionnaire elicited the following information: age class, gender, the price category of the next phone the respondent intended to purchase, and the





Notes: "The phone is of solid make and there is no need for its replacement for several years due to it breaking down; <sup>b</sup>the software, operating system and the phone components (e.g. screen or keyboard) are easy to update; <sup>c</sup>the manufacturer is responsible for the collection of the replaced phones for recycling; the metal and plastic parts are recycled at the minimum; possibly part of the replaced mobiles are refurbished for new users; <sup>d</sup>the hazardous materials are handled with special care to guarantee safety and there amount is minimized; the waste is minimized and recycled; <sup>e</sup>the manufacturer supervises its vendor related to the working environment, wage levels and labor rights as well as is responsible that appropriate attention is devoted to environment control

frequency of changing phones. Five additional questions were included from the TNS Atlas survey of Finnish consumers[1]. The Atlas statements were measured on a scale ranging from one to five, with one referring to "I totally disagree" and five to "I totally agree". TNS Gallup confirmed that, according to their representative sample, age and gender affected attitudes towards sustainable development in Finland as follows: females are more concerned and the concern grows with age. Thus, these were natural descriptors in the questionnaire. We also wanted to find out how the frequency of purchasing a new mobile phone and the intended price class of the next phone affected choice preferences. Finally, we asked if the respondents paid for their phones themselves because we were interested in personal choice. The relevant TNS Gallup questions were:

- Atlas 1. I would accept a lower standard of living if it contributed to decreasing environmental pollution.
- Atlas 2. I would happily pay a somewhat higher price for an environmentally friendly product.
- · Atlas 3. I am prepared to change my way of life in order to protect the environment.



Figure 1. Example of a task in a conjoint questionnaire

391

IJPDLM 43,5/6

392

- *Atlas 4.* I try and avoid the purchase and use of products that pollute the environment.
- Atlas 5. There is too much fuss about environmental topics.

Our sample distributions of responses to the five Atlas questions all differed significantly from the distributions of the actual sample of the Atlas survey (risk level = 0.05,  $\chi^2$ -test). The consumption attitudes in our data clearly favored more sustainable development. The members' education was higher in our sample, and their ages younger on average than in the country-level sample. Thus, the effect of education on attitudes to green consumption were stronger compared with the country-level sample than the effect of the younger age, which worked in the opposite direction (TNS Gallup Finland, 2012, personal communication (9 August)).

We also found that, measured on the Atlas questions, the female respondents in our sample had a more favorable attitude to green consumption (p < 0.0001 for all the Atlas questions), as did respondents over the age of 26 (p < 0.0001 except for Atlas 5, p = 0.09) compared with the younger age groups.

#### 5.2 The segments and their value functions

We estimated the value functions segment-wise using Latent Class clustering. Although in the aggregate model the two attribute interactions were significant, in the chosen cluster model they were not and thus we only used the segment-wise model with the attribute's main effects. The four- and five-cluster solutions had the lowest CAIC measures. We chose the four-cluster solution because it was easy to interpret and each of the clusters was of good size, whereas in the five-cluster solution the size of one of the clusters was only 3.5 percent. To each respondent is attached a probability (membership) of belonging to each cluster, and for each one the maximum membership is the largest membership over all clusters. The average maximum membership in the four-cluster solution was 0.90. The entropy measure, which is used to "assesses the fuzziness in membership" (Ramaswamy *et al.*, 1993), was 0.81 (one being the best and zero the worst value). The two tasks that were the same for all respondents were left for holdout (excluded in the estimation). The segment-wise value functions confirm that 68.5 percent of the respondent choices were actually made in the holdout tasks. All the value-function part-worths in the four estimated clusters were significant (p < 0.0001).

Table I shows the importance of the attributes across the clusters as well as the cluster sizes, rather than the part-worths of the attribute levels. For each attribute, the importance is the range of its part-worths divided by the sum of the ranges of all the attributes. We had only two possible levels for each attribute, and thus the importance scores are sufficient to describe the preferences in terms of a value function. The cluster size is the sum of the respondents' membership probabilities in the cluster.

We called the clusters updater, budgeter, environmentalist, and long-life user. Actually, physical strength and length of life is clearly the only attribute exceeding an importance of 10 percent in all of the clusters, and in other attributes the preferences differ a great deal across the clusters.

The updater appreciates software-updating possibilities in particular, but also values hardware updating more highly than the rest of the clusters. Males are more strongly represented in this cluster (p < 0.001), and relatively more members are under 27 than in the rest of the sample. Updaters are more willing to spend in excess of 300 euros on their



Attribute importance	Updater	Budgeter	Environ-mentalist	Long-life user	Preferences for sustainability
Physical strength and length of life	20.2	21.4	14.9	44.3	
Updating characteristics: software	30.5	8.5	6.1	10.6	
Updating characteristics: hardware	9.7	5.1	3.5	4.5	
Recycling (plastic and metal parts and					
phone recycling)	11.1	7.4	17.6	8.9	393
Hazardous materials and waste					
processing in production	8.3	8.7	24.6	8.7	
Ethical aspects and environmental					
conditions among the suppliers	8.4	7.2	27.6	8.9	Table I.
Price	11.8	41.7	5.7	14.1	Attribute importance
Total	100	100	100	100	across the clusters (%)
Cluster size	22.1	10.9	40.8	26.2	and the cluster sizes (%)

new mobiles (p < 0.001), and show a stronger tendency to replace them at intervals of 18 months at the most than the rest of the sample.

The budgeter sets a high priority on the current price (with no 10 percent premium) in his or her choice of mobile phone. Measured on the Atlas questions, attitudes toward green consumption are clearly less positive than among the other remaining clusters (all questions p < 0.001). The cluster members are more often males (p = 0.029), and they are more likely to buy phones costing 300 euros or less than the rest of the sample. Again on the Atlas questions, budgeters have fewer green values (p < 0.001) than the rest of the sample.

The environmentalist especially values suppliers that take ethical labor aspects and environmental effects into account, and attaches significance to the usage of hazardous materials in manufacturing as well as waste-processing practices. Likewise, this cluster emphasizes recycling more heavily than the rest of the sample. Environmentalists are more often females than males (p < 0.001), and more often 27 or older (p < 0.001), and they intend to buy phones priced at 300 euros or less (p = 0.006) than the rest of the clusters. Measured on the Atlas questions and compared with the rest of the sample, they also have greener values (p < 0.001).

The long-life user places high importance on the durability and long life of the phone (44.3 percent). A possible price increase of 10 percent is also unpleasant for this cluster, but much less so than for the budgeter: the long-life user is more often under 27 years of age (p = 0.041), has more often intentions to buy phones costing 300 euros or less (p < 0.001) than the others (Table II).

Note that the time interval for changing phones in this data exceeds 24 months for the great majority of the respondents: only 13 percent keep the same phone for 18 months or less.

#### 5.3 The validity, reliability, and limitations of the study

The focus group discussions followed a detailed but flexible schedule and incorporated free idea generation and discussion as well as evaluation of the statements in the four-theme framework. The following measures were taken in order to ensure the reliability of the results: all the discussions were recorded and transcribed (verbatim), with the permission of the interviewees; the same postdoctoral researcher was the moderator; and the same schedule was followed. The question of sustainable



IIDDI M						
43.5/6	Question	Updater	Budgeter	Environ-mentalist	Long-life user	All
20,070	Gender (%)					
	Male	30.4	11.0	32.4	26.2	60.7
	Female	11.3	7.5	55.6	25.6	39.3
	Age (%) (years)					
394	15-23	46.3	33.2	43.3	45.5	40.0
	24-26	27.5	24.4	27.0	25.4	25.9
	27-29	9.1	14.5	12.4	9.0	12.2
	30-49	16.6	24.3	16.5	15.7	19.7
	50-69	0.6	3.4	0.8	4.5	2.2
	Atlas questions					
	Atlas 1, mean	3.0	2.1	2.7	3.1	2.6
	Atlas 2, mean	2.5	1.8	2.4	3.1	2.2
	Atlas 3, mean	2.5	1.8	2.2	2.8	2.1
	Atlas 4, mean	2.8	2.2	2.5	3.0	2.5
	Atlas 5, mean	3.8	4.4	4.0	3.5	4.1
	Next phone: inten	ded price class	(%) (€)			
	Below 100	4.1	15.7	16.3	17.9	13.4
	101-200	15.3	29.4	32.0	28.4	26.8
	201-300	18.1	17.6	18.5	20.9	18.2
	301-400	18.8	13.3	14.3	9.7	14.5
	More than 400	43.75	24.1	19.01	23.13	27.18
	Time interval for	changing phor	ies (%) (months,	)		
	1-12	3.8	1.6	2.2	2.2	2.3
Table II.	13-18	13.1	10.2	9.1	11.9	10.7
Descriptors across	19-24	28.4	19.1	21.5	20.9	22.0
the clusters	More than 24	54.7	69.2	67.2	64.9	65.0

development in the context of mobile phones attracted considerable interest in the groups and was dealt with thoroughly, thereby enhancing the validity of the results.

The reliability of the CA is discussed in Section 5.2. The fit of the cluster results was very good, and even the holdout tasks were predicted with high probability. Starting from the focus-group results the elaboration towards the final determination of the attributes and levels proceeded via two additional pretests (quantitatively by means of a questionnaire, and qualitatively in further individual interviews). The final choice of design was thus a lengthy process in which the features of sustainability were reduced to five in order to keep the conjoint-analysis tasks cognitively simple enough. Several steps were taken in order to guarantee a conjoint design with the most significant attributes and thereby ensure validity. We have already acknowledged the need to view the price attribute with special care. Additionally, the anonymity of the web questionnaire helped to minimize social pressures and thus to encourage genuine responses.

As already pointed out, the attitudes of the sample population were much more positive towards green consumption as measured on the Atlas questions compared with the general Finnish population. Throughout we would at least expect the environmentalist cluster to be smaller, and the budgeter cluster to be larger in a representative sample of our target group.

Our results indicate that there are mobile-phone users who are prepared to pay at least 10 percent more for improved sustainability features. This finding is in harmony with the quantitative pre-tests and the focus-group discussions. However, hypothetical



methods such as CBC and open-ended questioning have been criticized on the grounds that their reliability in the measurement of willingness to pay is biased. Miller *et al.* (2011) include an excellent list of references and find evidence of an upwards bias in CBC purchase-probability results. We refer in Section 3 to some studies employing scanner panel data, and others involving real transactions against money, suggesting that there is a willingness to pay premiums for sustainable features.

#### 6. Potential implications for supply chains and supply chain management

It is likely that sustainability issues will carry more and more weight in customers' product-choice decisions in the future (Schischke *et al.*, 2005). Market studies have already provided evidence that values are shifting towards sustainability: the majority of consumers prefer brands that are environmentally and socially responsible (Kleanthous and Peck, 2006). Earlier research also confirms that there is a group of consumers who are willing to pay a premium for environmentally friendly products (Laroche *et al.*, 2001). Our focus in this section is on the potential implications of these developments for SCs and SCM as evidenced in our results. The implications extend to sustainable product design, also called eco-design, one of the benefits of which is that it changes the perspective on the product and could trigger new, highly innovative concepts (Schischke *et al.*, 2005). In order to survive in today's competitive business environment, companies need to provide highly innovative and value added products of high quality, and bring them to market quickly and effectively (Hilletoft *et al.*, 2010).

Our findings highlight at least four significant issues. The first is related to the prolonged expected lifecycle of the mobile phone; the second and third concern the effects of improved updating characteristics on product design and manufacturing, and on sales channels, and the fourth is related to the environmental and social aspects in the whole SC.

First, the issue of a prolonged expected lifecycle for mobile phones arose in all the focus-group discussions and was considered important. In the conjoint study we also found a respondent cluster that emphasized the importance of durability and a longer life. This is a highly significant finding, seemingly directly opposing current trends in the market that clearly promote shorter life cycles. According to Zadok and Puustinen (2010), the average mobile-phone replacement rate is about 18 months, which is the highest so far in the history of consumer electronic devices. The trend of shortening life cycles is evident across the whole market for such devices, which affects their end-of-life management. We could therefore ask what problematic effects the shortening life cycles would have on sustainable development. Durability and the longer life cycle that were highlighted in our results have market and product-design implications, potentially leading to a re-shaping of the mobile-phone product portfolio. The market implication is the potential effect on the growth in sales volumes. This could lead to reductions in raw-material consumption, manufacturing volumes, and the number of phones to be recycled. Manufacturers may thus need to consider designs that have more timeless features, and to find materials, either new or currently available, that support a longer life cycle and end-of-life recycling. This, in turn, could lead to new innovative design and materials solutions.

The second issue is related to the potential effects of updating characteristics on the design and manufacturing of mobile phones, and also to their durability and longer life cycles. We asked the respondents to rate the relative importance of updating software



Preferences for sustainability

and hardware, of which the updating of software was considered more important on average. The possible implication here is that manufacturers could consider the feasibility of updating and loading new software applications to a mobile phone over a longer time frame than is currently possible. The updating of hardware, on the other hand, might affect the product-design phase and could even radically change customer offerings in the market, extending the component-modularity aspect from manufacturing to the customer. Consumers currently choose their mobile phone from a set of available models, and in most cases there is no opportunity to (mass)customize the final product. If consumers value updating possibilities, manufacturers seeking competitive advantage should develop more modular product designs. It is possible that customized modular mobile phones will be available in the coming years. Moreover, mobile services carry great promise for the future (Lyytinen et al., 2004), thereby assigning mobile phones an increasingly essential role in everyday life regardless of whether they are closer to tablets or are more traditional in design. The customer could build up his or her phone from modules as if they were Lego bricks, and later on replace some elements such as the screen or camera with more modern versions.

The third issue concerns the effects of the improved updating characteristics on sales channels. Modularity is an accepted feature of mobile phones, and thus is familiar in the upstream supply-chain phases such as product manufacturing. Modular products are typically constituted of different parts, and many companies seek opportunities for "mixing and matching", i.e. using the same modules (standard or of their own design) in products within the same product family, or even across families. This is also the case with mobile phones. However, in its current form, from the users' perspective this means offering exclusively fixed products, and there are virtually almost no customization options at the customer interface (downstream of the SC). Service modularity is a new and emerging research approach (Voss and Hsuan, 2009; Starr, 2010), the starting phase being at the customer interface (Bask et al., 2011). In the case of mobile phones we take two examples of modularity and customization that affect the offering at the customer interface. First, modularity there could be predetermined options allowing slight customization ("mix and match"; see, e.g. NikeID, Smart car) in the form of sustainable and non-sustainable modules, for example. Thus, customers could decide on the degree of sustainability in their choice of mobile phone. Second, modularity and customization could be exploited after the sale, during the usage stage. The life cycle of the phone could be extended through the upgrading of modules during the usage cycle. Thus, a modular structure enables product upgrading through the life-cycle (Brusoni and Prencipe, 2001). The effect will be to increase modular choice options in retailing channels, and therefore the implications are significant.

The focus in the fourth issue is on the environmental and social implications in the whole SC. It is worth remembering that about 80 percent of all product-related environmental impacts and life-cycle costs are determined during the product-design phase (Schischke *et al.*, 2005). The first step towards sustainable design could be to consider the manufacturing costs. Minimizing the material throughput per product unit will reduce costs and make the product "greener" (Schischke *et al.*, 2005). Current challenges in green designs include the lack of global sustainability standards and cost-effective manufacturing processes (Zadok and Puustinen, 2010). The next step is to integrate thoroughly the life-cycle design, including recycling aspects, into the product design. Members of the environmentalist cluster valued the recycling of plastic and metal parts and used phones, for example. They also felt strongly that hazardous



IJPDLM

43.5/6

materials should not be used in the production process, and supported efforts to increase waste processing. Furthermore, their concern about ethical and environmental matters among suppliers was strong. The question remains of how to make these sustainability efforts of mobile phone companies visible to consumers.

Finally, our results show that there are consumers who are willing to pay more for a sustainable product, although we only accounted for a price increase of as low as 10 percent in our conjoint questionnaire. The willingness to pay more also has implications for SCs related to the use of more sustainable (and possibly more expensive) materials in mobile-phone production, for example. If companies were convinced that there was a consumer segment willing to pay a premium for a green phone they would be prepared to promote such a product.

#### 7. Conclusions

Using customer-focus-group discussions we identified the most important attributes related to the sustainability of mobile phones. These discussions were effective in defining the perceived key features, some of which turned out to be unexpected. By means of CBC analysis we measured the preferences related to the product attributes as well as to the price attribute. On the basis of the CBC results we found segments with different preferences, which we named updater, budgeter, environmentalist, and long-life user. These segments enabled us to identify several potential classes of SCM implications. The first of these concerned the prolonged expected lifecycle of the mobile phone. The second emerged from the preference for improved updating characteristics, and the third was related to how such features could induce changes in sales channels. The fourth and final segment focused on the environmental and social aspects of the SC. All the implications seemed to affect sustainable product design, also called product eco-design. Moreover, we found that some consumers were willing to pay a premium for phones with improved sustainable features. We would like to emphasize the fact that physical strength and length of life constituted the only attribute that exceeded the importance rating of 10 percent across all of the clusters (being 14.9 percent at its lowest). This finding, combined with the perceived importance of updating characteristics, is highly interesting. Such preferences among consumers are in conflict with current market trends in which both a rapid increase in the number of mobile phones and a shortening life cycle are highly visible phenomena.

We also included in the CBC questionnaire five questions measuring attitudes towards sustainable development employed in TNS Atlas, a major survey of Finnish consumers. Comparison of the country level and our sample results led us to conclude that our respondents had a significantly more positive attitude towards green consumption. This could undoubtedly be attributed to their higher education, and we would expect at least a subset of the respondents to be innovators and early adopters of products and services with sustainable features. We also acknowledge that the segments we identified, although assumed generally valid among Finnish consumers paying for their phones form their own pockets, would differ in relative size from the segments in the target population.

In this study, we have brought to light some fundamental and perhaps unanticipated observations on customer attitudes towards the sustainability features of mobile phones, and have gauged customer preferences for these features. There seem to be no other studies that have attempted to identify the most relevant sustainable development



Preferences for sustainability features related to the whole SC from the consumer's point of view. We believe our results are of interest to manufacturers of mobile phones in the development of products and services. This has been confirmed by discussions with a major manufacturer that does consider the role of sustainability among other features, including price, in a consumer's choice of mobile phone. Our aim was rather to identify the different facets of sustainable development as well as to evaluate their relative importance – a task not undertaken in any other studies. More precisely, companies could use our results in their product development by designing phones with a better match to consumer preferences, customizing advertisement messages, and finding improved and novel ways of managing their SCs. We have noticed that several companies have strengthened their positions on issues related to sustainability, although consumers do not seem to be very conscious of this yet.

The results of this study and the potential implications for SCs and their management offer several future research avenues. First, our study takes the first steps in opening up the "black box" of customer preferences regarding sustainability features in a SC offering a choice of mobile phones. The preliminary results are promising and therefore it would be worthwhile extending the study to include other groups of mobile phone users, as well as other types of products and services. Second, the study identified a need for more in-depth research into the triple bottom line of SSCM, as Carter and Rogers (2008) and Carter and Easton (2011) suggest. In addition to pursuing environmental and social development goals, sustainable development should focus on improving companies' and SCs' long-term economic performance. Thus, efforts should be made to mass-market sustainable products in order to achieve commercial success and environmental sustainability (Dangelico and Pujari, 2010).

Moreover, there are strong signs that in the near future mobile phones will be intelligent consumer devices that are essential for accessing everyday services and are constantly updated with new applications. The entire ecosystem of marketing could change. For instance, companies could send fully customized marketing messages on the basis of geographical location, one could check the availability and prices of products and services in alternative retail locations, and the phone could be used as a paying device in everyday transactions. Thus, innovative technological modules could be available for the mobile phone.

#### Note

**IIPDLM** 

43,5/6

398

1. TNS Atlas is a major survey concerning Finnish consumer consumption, brand relationship and media use. The data gathering is continuous and its different parts have 4,000-20,000 respondents annually aged between 15 and 69 years. The sample is representative of Finnish inhabitants and the data gathering takes place on the internet.

#### References

- Andersen, M. and Skjoett-Larsen, T. (2009), "Corporate social responsibility in global supply chains", Supply Chain Management: An International Journal, Vol. 14 No. 2, pp. 75-86.
- Angell, L.C. and Klassen, R.D. (1999), "Integrating environmental issues into the mainstream: an agenda for research in operations management", *Journal of Operations Management*, Vol. 17 No. 5, pp. 575-598.
- Auger, P., Burke, P., Devinney, T.M. and Louviere, J.J. (2003), "What will consumers pay for social product features?", *Journal of Business Ethics*, Vol. 42, pp. 281-304.



Bask, A. and Kuula, M. (2011), "Measuring supply chain level environmental sustainability – case Nokia", *International Journal of Business Insights and Transformation*, Vol. 3 No. 3, pp. 16-24.

- Bask, A., Lipponen, M., Rajahonka, M. and Tinnilä, M. (2011), "Framework for modularity and customization: a service perspective", *Journal of Business & Industrial Marketing*, Vol. 26 No. 5, pp. 306-319.
- Bjorner, T.B., Hansen, L.G. and Russell, C.S. (2004), "Environmental labeling and consumers' choice – an empirical analysis of the effect of the Nordic Swan", *Journal of Environmental Economics and Management*, Vol. 47, pp. 411-434.
- Brusoni, S. and Prencipe, A. (2001), "Unpacking the black box of modularity: technologies, products and organizations", *Industrial and Corporate Change*, Vol. 10 No. 1, pp. 179-205.
- Carter, C.R. (2005), "Purchasing social responsibility and firm performance: the key mediating roles of organizational learning and supplier performance", *International Journal of Physical Distribution & Logistics Management*, Vol. 35 No. 3, pp. 177-194.
- Carter, C.R. and Easton, P.L. (2011), "Sustainable supply chain management: evolution and future directions", *International Journal of Physical Distribution & Logistics Management*, Vol. 41 No. 1, pp. 46-62.
- Carter, C.R. and Ellram, L.M. (1998), "Reverse logistics: a review of the literature and framework for future investigation", *Journal of Business Logistics*, Vol. 19 No. 1, pp. 85-102.
- Carter, C.R. and Jennings, M.M. (2002), "Logistics social responsibility: an integrative framework", *Journal of Business Logistics*, Vol. 23 No. 1, pp. 145-180.
- Carter, C.R. and Rogers, D.S. (2008), "A framework of sustainable supply chain management: moving toward new theory", *International Journal of Physical Distribution & Logistics Management*, Vol. 38 No. 5, pp. 360-387.
- Chan, R.Y.K. (2001), "Determinants of Chinese consumers' green purchase behavior", *Psychology & Marketing*, Vol. 18 No. 4, pp. 389-413.
- Chen, C. (2001), "Design for the environment: a quality-based model for green product development", *Management Science*, Vol. 47 No. 2, pp. 250-263.
- Chen, S. and Bouvain, P. (2009), "Is corporate responsibility converging? A comparison of corporate responsibility reporting in the USA, UK, Australia, and Germany", *Journal of Business Ethics*, Vol. 87, pp. 299-317.
- Chrzan, K. and Orme, B. (2000), An Overview and Comparison of Design Strategies for Choice Based Conjoint Analysis, available at: www.sawtoothsoftware.com/download/techpap/ desgncbc.pdf (accessed 16 December 2011).
- Dangelico, R.M. and Pujari, D. (2010), "Mainstreaming green product innovation: why and how companies integrate environmental sustainability", *Journal of Business Ethics*, Vol. 95, pp. 471-486.
- Dangelio, R.M. and Pontrandolfo, P. (2010), "From green product definitions and classifications to Green Option Matrix", *Journal of Cleaner Production*, Vol. 18, pp. 1608-1628.
- Darnall, N., Jolley, G.J. and Handfield, R. (2008), "Environmental management systems and green supply chain management: complements for sustainability?", *Business Strategy and the Environment*, Vol. 17 No. 1, pp. 30-45.
- Daugherty, P.J., Autry, C.W. and Ellinger, A.E. (2001), "Reverse logistics: the relationship between resource commitment and program performance", *Journal of Business Logistics*, Vol. 22 No. 1, pp. 107-123.



399

Preferences for

sustainability

IJPDLM 43,5/6	Defee, C.C., Esper, T. and Mollenkopf, D. (2009), "Leveraging closed-loop orientation and leadership for environmental sustainability", <i>Supply Chain Management: An International</i> <i>Journal</i> , Vol. 14 No. 2, pp. 87-98.
	DeSarbo, W.S., Ramaswamy, V. and Cohen, S.H. (1995), "Market segmentation with choice-based conjoint analysis", <i>Marketing Letters</i> , Vol. 6 No. 2, pp. 137-147.
400	<ul> <li>Eggers, F. and Eggers, F. (2011), "Where have all the flowers gone? Forecasting green trends in the automobile industry with a choice-based conjoint adoption model", <i>Technological Forecasting &amp; Social Change</i>, Vol. 78, pp. 51-62.</li> </ul>
	Fabbe-Costes, N., Roussat, C. and Colin, J. (2011), "Future sustainable supply chains: what should companies scan?", <i>International Journal of Physical Distribution &amp; Logistics Management</i> , Vol. 41 No. 3, pp. 228-252.
	Fiksel, J. (2001), "Measuring sustainability in ecodesign", in Charter, M. and Tischner, U. (Eds), Sustainable Solutions – Developing Products and Services for the Future, Greenleaf, Sheffield, pp. 165-187.
	Finn, A. and Louviere, J.J. (1992), "Determining the appropriate response to evidence of public concerns: the case of food safety", <i>Journal of Public Policy &amp; Marketing</i> , Vol. 11 No. 1, pp. 12-25.
	Gartner (2009), Press release from 3 February, available at: www.gartner.com/it/page.jsp? id=904729 (accessed 5 September 2011).
	Gibbs, A. (1997), "Focus groups", Social Research Update, Department of Sociology, University of Surrey, Guildford, pp. 1-8.
	GSMA (2010), Annual Review, 2 February 2011, available at: http://gsmworld.com/documents/ PPAR_2011_1_Mar.pdf
	Guide, V.D.R. Jr (2000), "Production planning and control for remanufacturing: industry practice and research needs", <i>Journal of Operations Management</i> , Vol. 18, pp. 467-483.
	Guide, V.D.R. Jr and van Wassenhove, L.N. (2002), "The reverse supply chain: smart manufacturers are designing efficient processes for reusing their products", <i>Harvara</i> <i>Business Review</i> , Vol. 8 No. 2, pp. 25-26.
	Halcomb, E.J., Gholizadeh, L., DiGiacomo, M., Phillips, J. and Davidson, P.M. (2007), "Literature review: considerations in undertaking focus group research with culturally and linguistically diverse groups", <i>Journal of Clinical Nursing</i> , Vol. 16 No. 6, pp. 1000-1011.
	Handfield, R., Sroufe, R. and Walton, S. (2005), "Integrating environmental management and supply chain strategies", <i>Business Strategy and the Environment</i> , Vol. 14, pp. 1-19.
	Hearne, R.R. and Salinas, Z.M. (2002), "The use of choice experiments in the analysis of tourist preferences for ecotourism development in Costa Rica", <i>Journal of Environmental</i> <i>Management</i> , Vol. 65, pp. 153-163.
	Hervani, A.A., Helms, M.M. and Sarkis, J. (2005), "Performance measurement for green supply chain management", <i>Benchmarking: An International Journal</i> , Vol. 12 No. 4, pp. 330-353.
	Hilletoft, P., Ericsson, D. and Lumsten, K. (2010), "Coordinating new product development and supply chain management", <i>International Journal of Value Chain Management</i> , Vol. 4 Nos 1/2, pp. 170-192.
	Hughner, R.S., McDonagh, P., Prothero, A., Shultz, C.J. II and Stanton, J. (2007), "Who are organic food consumers? A compilation and review of why people purchase organic food", <i>Journal</i> of Consumer Behavior, Vol. 6 Nos 2/3, pp. 94-110.
	Intuit 2020 Report (2010), <i>The Future of Financial Services</i> , available at: http://http- download.intuit.com/http.intuit/CMO/intuit/futureofsmallbusiness/intuit_corp_banking. pdf (accessed 6 August 2012).
	•• .) •[]
للاستشاران	

#### www.man

- Kaplan, W. (2006), "Can the ubiquitous power of mobile phones be used to improve health outcomes in developing countries?", *Globalization and Health*, Vol. 2 No. 1, p. 9.
- Karniouchina, E.V., Moore, W.L., van der Rhee, B. and Verma, R. (2009), "Issues in the use of rating-based versus choice-based conjoint analysis in operations management", *European Journal of Operational Research*, Vol. 197, pp. 340-348.
- Kitzinger, J. (1995), "Qualitative research: introducing focus groups", British Medical Journal, Vol. 311, pp. 299-302.
- Kleanthous, A. and Peck, J. (2006), Let Them Eat Cake: Satisfying the New Consumer Appetite for Responsible Brands, WWF, available at: www.wwf.org.uk/filelibrary/pdf/let\_them\_eat\_ cake\_abridged.pdf
- Kleindorfer, P.R., Singhal, K. and van Wassenhove, L.N. (2005), "Sustainable operations management", *Production and Operations Management*, Vol. 14 No. 4, pp. 482-492.
- Krueger, R.A. and Casey, M.A. (2000), Focus Groups: A Practical Guide for Applied Research, 3rd ed., Sage, Thousand. Oaks, CA, p. 215.
- Laroche, M., Bergeron, J. and Barbaro-Forleo, G. (2001), "Targeting consumers who are willing to pay more for environmentally friendly products", *Journal of Consumer Marketing*, Vol. 18 No. 6, pp. 503-520.
- Lee, K-H. and Kim, J.-W. (2009), "Current status of CSR in the realm of supply management: the rise of the Korean electronics industry", *Supply Chain Management: An International Journal*, Vol. 14 No. 2, pp. 138-148.
- Linton, J.D., Klassen, R. and Jayaraman, V. (2007), "Sustainable supply chains: an introduction", Journal of Operations Management, Vol. 25, pp. 1075-1082.
- Luchs, M.G., Walker Naylor, R., Irwin, J.R. and Raghunathan, R. (2010), "The sustainability liability: potential negative effects of ethicality on product preference", *Journal of Marketing*, Vol. 74 No. 5, pp. 18-31.
- Luttropp, C. and Lagerstedt, J. (2006), "EcoDesign and the ten golden rules: generic advice for merging environmental aspects into product development", *Journal of Cleaner Production*, Vol. 14, pp. 1396-1408.
- Lynn, J. (2010), "Mobile phones help lift poor out of poverty: UN study", Reuters Prints, Thomson Reuters, Mylapore.
- Lyytinen, K., Upkar, V., Ackerman, M.S., Davis, G., Avital, M., Robey, D., Sawyer, S. and Sorensen, C. (2004), "Surfing the next wave: design and implementation challenges of ubiquitous computing environments", *Communications of the Association for Information Systems*, Vol. 13, pp. 697-716.
- McFadden, D. (1974), "Conditional logit analysis of qualitative choice behavior", in Zarembka, P. (Ed.), *Frontiers in Econometrics*, Academic Press, New York, NY, pp. 105-142.
- Michaud, C., Llerena, C. and Joly, I. (2012), "Willingness to pay for environmental attributes of non-food products a real choice experiment", Working Paper of Laboratoire d'Economie Appliquee de Grenoble.
- Miller, K.M., Hofstetter, R., Krohmer, H. and Zhang, J. (2011), "How should consumers' willingness to pay be measured? An empirical comparison of state-of-the-art approaches", *Journal of Marketing Research*, Vol. 48 No. 1, pp. 172-184.
- Moon, W., Florkowski, W.J., Bruckner, B. and Schonhof, I. (2002), "Willingness to pay for environmental practices: implications for eco-labeling", *Land Economics*, Vol. 78 No. 1, pp. 88-102.

Morgan, D.L. (1996), "Focus groups", *Annual Review of Sociology*, Vol. 22, pp. 129-152. Morgan, D.L. (1998), *Focus Groups as Qualitative Research*, Sage, Thousand Oaks, CA.



Preferences for sustainability

401

IJPDLM	Nidumolu, R., Prahalad, C.K. and Rangaswami, M.R. (2009), "Why sustainability is now the key driver of innovation", <i>Harvard Business Review</i> , Vol. 87 No. 9, pp. 56-64.
±0,0/U	Potoglou, D. and Kanaroglou, P.S. (2007), "Household demand and willingness to pay for clean vehicles", <i>Transportation Research Part D: Transport and Environment</i> , Vol. 12 No. 4, pp. 264-274.
402	<ul> <li>Ramaswamy, V., DeSarbo, W.S., Reibstein, D.J. and Robinson, W.T. (1993), "An empirical pooling approach for estimating marketing mix elasticities with PIMS data", <i>Marketing Science</i>, Vol. 12 No. 1, pp. 103-124.</li> </ul>
	Reutterer, T. and Kotzab, H.W. (2000), "The use of conjoint-analysis for measuring preferences in supply chain design", <i>Industrial Marketing Management</i> , Vol. 29 No. 1, pp. 27-35.
	Roe, B., Teisl, M.F., Levy, A. and Russell, M. (2001), "US consumers' willingness to pay for green electricity", <i>Energy Policy</i> , Vol. 29, pp. 917-925.
	Rogers, D.S. and Tibben-Lembke, R. (2001), "An examination of reverse logistics practices", Journal of Business Ethics, Vol. 22 No. 2, pp. 129-148.
	Rogers, E.M. (1962), Diffusion of Innovations, The Free Press, Glencoe.
	Rogers, E.M. (1976), "New product adoption and diffusion", <i>Journal of Consumer Research</i> , Vol. 2 No. 4, pp. 290-301.
	Rokka, J. and Uusitalo, L. (2008), "Preference for green packaging in consumer product choices – do consumers care?", <i>International Journal of Consumer Studies</i> , Vol. 32, pp. 516-525.
	Sammer, K. and Wustenhagen, R. (2006), "The influence of eco-labelling on consumer behaviour – results of a discrete choice analysis for washing machines", <i>Business Strategy and the</i> <i>Environment</i> , Vol. 15, pp. 185-199.
	Schischke, K., Hagelüken, M. and Steffenhagen, G. (2005), An Introduction to Ecodesign Strategies – Why, What and How?, available at: www.ecodesignarc.info/servlet/is/203/EN_An% 20Introduction%20to%20EcoDesign%20Strategies.pdf?command=downloadContent& filename=EN_An%20Introduction%20to%20EcoDesign%20Strategies.pdf
	Scott, J.T. (2008), <i>Managing the New Frontiers: An Introduction to the Fundamentals</i> , Management Education Services, Panama City, FL.
	Seuring, S. and Muller, M. (2008), "From a literature review to conceptual framework for sustainable supply chain management", <i>Journal of Cleaner Production</i> , Vol. 16, pp. 1699-1710.
	Shang, K.C., Lu, C.S. and Li, S. (2010), "A taxonomy of green supply chain management capability among electronics-related manufacturing firms in Taiwan", <i>Journal of</i> <i>Environmental Management</i> , Vol. 91, pp. 1218-1226.
	Shiloy, A. (2009), "Few consumers willing to pay more for environmentally friendly mobile phones", available at: www.xbitlabs.com/news/mobile/display/20091019100003_Few_ Consumers_Willing_to_Pay_More_for_Environmentally_Friendly_Mobile_Phones.html (accessed 28 June 2012).
	Sonnenberg, N.D., Erasmus, A.C. and Donoghue, S. (2011), "Significance of environmental sustainability issues in consumers' choice of major household appliances in South Africa", <i>International Journal of consumer Studies</i> , Vol. 35, pp. 153-163.
	Spence, L. and Bourlakis, M. (2009), "The evolution from corporate social responsibility to supply chain responsibility: the case of Waitrose", <i>Supply Chain Management: An International</i> <i>Journal</i> , Vol. 14 No. 4, pp. 291-302.
	Srivastava, S.K. (2008), "Network design for reverse logistics", The International Journal of Management Science, Vol. 36, pp. 535-548.
اللاستشاران	المنارة

Stahel, W.R. (2001), "Sustainability and services", in Charter, M. and Tischner, U. (Eds), Sustainable Solutions – Developing Products and Services for the Future, Greenleaf, Sheffield, pp. 151-164.

- Stank, T.P., Dittman, J.P. and Autry, C.W. (2011), "The new supply chain agenda: a synopsis and directions for future research", *International Journal of Physical Distribution & Logistics Management*, Vol. 41 No. 10, pp. 940-955.
- Starr, M.K. (2010), "Modular production a 45-year-old concept", International Journal of Operations & Production Management, Vol. 30 No. 1, pp. 7-19.
- Stock, J.R. (1998), Development and Implementation of Reverse Logistics Programs, Council of Logistics Management, Oak Brook, IL.
- Swait, J. and Louviere, J.J. (1993), "The role of the scale parameter in the comparison and estimation of multinomial logit models", *Journal of Marketing Research*, Vol. 30, August, pp. 305-314.
- The Sustainability Yearbook (2012), available at: www.sustainability-indexes.com/images/samyearbook-2012-final\_tcm1071-337504.pdf (accessed 29 June 2012).
- Tischner, U. (2001), "Tools for ecodesign and sustainable product design", in Charter, M. and Tischner, U. (Eds), *Sustainable Solutions – Developing Products and Services for the Future*, Greenleaf, Sheffield.
- Ubilava, D., Foster, K.A., Lusk, J.L. and Nilsson, T. (2010), "Effects of income and social awareness on consumer WTP for social product attributes", *Technological Forecasting & Social Change*, Vol. 77, pp. 587-593.
- van Doorn, J. and Verhoef, P.C. (2011), "Willingness to pay for organic products: differences between virtue and vice foods", *International Journal of Research in Marketing*, Vol. 28, pp. 167-180.
- Veisten, K. (2007), "Willingness to pay for eco-labelled wood furniture: choice-based conjoint analysis versus open-ended contingent valuation", *Journal of Forest Economics*, Vol. 13, pp. 29-48.
- Voss, C.A. and Hsuan, J. (2009), "Service architecture and modularity", *Decision Sciences*, Vol. 40 No. 3, pp. 541-569.
- Zadok, G. and Puustinen, R. (2010), "The green switch: designing for sustainability in mobile computing", *Proceedings of the First USENIX Sustainable IT Workshop in February* (*SustainIT 2010*), available at: www.thegreenswitch.org/wp content/uploads/2009/12/ TheGreenSwitch\_publication\_Jan2010.pdf (accessed 5 September 2011).

#### Further reading

SSI Web v6.0 (2008), "Sawtooth software", Technical Paper Series, available at: www. sawtoothsoftware.com/download/techpap/cbctech.pdf (accessed 2 November 2011).

(The Appendix follows overleaf.)



Preferences for sustainability

# 403

# IJPDLMAppendix. Statements regarding four-theme sustainability framework43,5/6

# 404

	Sustainable strategy and policy	1	2	3
1	I find it important that the mobile phone manufacturer has defined and published sustainability strategy			
2	I find it important that the mobile phone manufacturer regularly publishes a corporate social responsibility (CSR) report			
3	<ul> <li>I find it important that the mobile phone manufacturer can be trusted to provide information about the sustainability of the mobile phone during its life cycle: for example, that the manufacturer has an environmental life-cycle-management tool in place (such as EMS, LCA). This would imply that the information covered, for example;</li> <li>the product's energy and material use as well as emissions</li> <li>evaluations of possible effects on the environment</li> <li>plans for utilizing the results in decisions related to product design</li> </ul>			
4	I find it important that the mobile phone manufacturer's environmental strategy includes waste, energy and emission issues.			
5	I find it important that the manufacturer is interested in societal development and participates actively in international programs promoting sustainable development			
6	I find it important that the manufacturer supports mobile services focused on society as a whole			

	Sustainable product design	1	2	3
1	I find it important that renewable resources are favored in the design of mobile phones			
2	I find it important that recycling and disassembly (dismantling) are taken into account in the design process in order to facilitate recycling and the reuse of recyclable materials and components.			
3	I find it important that in the design of mobile phones the manufacturer imposes more strict requirements related to environmentally hazardous materials than EU regulations demand, for example.			
4	I find it important that the mobile phone manufacturer uses a third-party certified analytical tool in the product-design phase			
5	I find it important that mobile phone has a long life. For example, that it is possible to upgrade or update the phone (hardware and software) as well as to modify it according to the user's needs.			

(continued)



	Sustainable sourcing	1	2	3	Preferences for
1	I find it important that the mobile phone manufacturer can reliably evaluate (e.g., by means of a certification tool) its suppliers in terms of sustainability.				Sustamability
2	I find it important that the mobile phone manufacturer gives clear instructions for its own and its suppliers' use of resources and waste, and that it regularly monitors its suppliers.				405
3	I find it important that the mobile phone manufacturer regularly audits that its first-tier suppliers (those with a direct contract with the manufacturer) follow the environmental policies, procedures and practices to which they have committed, and develop these actions if required.				403
4	I find it important that the mobile phone manufacturers trains its suppliers in environmental issues				*
5	I find it important that the mobile phone manufacturer systematically shares information about materials that require control (including high-risk and hazardous materials)				*
6	I find it important that the mobile phone manufacturer aims at minimizing transportation, and pays attention to sustainability in transportation and packaging. For example, sustainability should be taken into account in the selection of transportation mode and materials. On the other hand, the product's packaging (manuals etc.) should be as efficient as possible.				
7	I find it important that the mobile phone manufacturer's working conditions follow common ethical principles.				*

	Product recovery and en-of-life managementd	1	2	3
1	I find it important that the mobile phone manufacturer follows more strict practices in its reuse and recycling of materials than required by the EU			
2	I find it important that the mobile phone manufacturer actively participates in programs promoting product recycling			
3	I find it important that the mobile phone manufacturer re-sells returned and unsold products, or repairs and up-grades them.			
4	I find it important that the recycling of mobile phones follows the principles of sustainable development. For example the manufacturer should clearly indicate how recycled materials and components are intended to be re-used			

#### About the authors

Dr Anu Bask is an Academy of Finland Postdoctoral Researcher in the Department of Information and Service Economy at the Aalto University School of Business in Finland. She is Director of the Kataja's Finnish Graduate School of Logistics and Supply Chain Management. She has published several articles in international refereed journals. Her research interests include supply chain management, supply chain relationships, business models, modularity, services, service triads, service processes, and sustainable supply chains. She has also been a reviewer for several international journals and conferences, and a co-editor of a special issue of *The International Journal of Productivity and Performance Management*. Anu Bask is the corresponding author and can be contacted at: anu.bask@aalto.fi

Dr Merja Halme is Professor of Management Science in the Department of Information and Service Economy at the Aalto University School of Business in Finland. Her research interests



IJPDLM 43,5/6	consist of business analytics, efficiency analyses and optimization. She has published articles in leading international journals on management science and operations research. Dr Markku Kallio is Professor of Management Science in the Department of Information and Service Economy at the Aalto University School of Business in Finland. He holds a Ph.D from Stanford University. His research interests include optimization and its applications in logistics, finance, marketing and economics. He has published articles in leading international journals on
406	<ul> <li>management science and operations research.</li> <li>Dr Markku Kuula is Professor of Logistics in the Department of Information and Service</li> <li>Economy at the Aalto University School of Business in Finland and current president of the Finnish Operations Research Society. He has long experience of assignment of chief information officer at Helsinki School of Economics (1996-2009). Before that he has worked as assistant professor in the Management Sciences (1994-1996). He has published in several refereed international journals and acted as member of organization committee for example at EURO/TIMS conference. His main research interests are multi-criteria selection problems of flexible technologies and different kinds of negotiation problems as well as investment decisions</li> </ul>

supply chain sustainability.

(for example taking environmental issues into consideration), industrial benchmarking and

To purchase reprints of this article please e-mail: **reprints@emeraldinsight.com** Or visit our web site for further details: **www.emeraldinsight.com/reprints** 

![](_page_26_Picture_2.jpeg)

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

![](_page_27_Picture_1.jpeg)